Author's response to reviews

Title: Practice nurse-based, individual and video-assisted patient education in oral anticoagulation. Protocol of a cluster-randomized controlled trial

Authors:

Thanh Duc Hua (thanhduc.hua@med.uni-goettingen.de)
Stefan Viktor Vormfelde (vormfelde@medizin.uni-goettingen.de)
Manar Abu Abed (manar.abuabed@med.uni-goettingen.de)
Hannelore Schneider-Rudt (hschnei4@gwdg.de)
Petra Sobotta (petra@sobotta-net.de)
Tim Friede (tim.friede@med.uni-goettingen.de)
Jean-François Chenot (jchenot@gwdg.de)

Version: 4 Date: 4 April 2011

Author's response to reviews:

Dear Editors,

Attached you will find our revised manuscript intended for publication in BMC Family Practice.

You will find the following changes in our revised manuscript:

Major revisions

1)References: References should be cited in the text using consecutive numbers in square brackets. Currently reference 16 is missing

We have cited all references in the text in square brackets. We corrected the reference list and cited them in the correct order. Reference No. 16 is not missing anymore.

2)Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

We checked our manuscript for typographical mistakes.

We hope that our revised manuscript comply with your entire expectations. Please contact us if there are any questions.

Yours sincerely

Dr Hua